Contents lists available at ScienceDirect





### Journal of International Management

journal homepage: www.elsevier.com/locate/intman

# International governance mode choice: Evidence from Brazilian franchisors

Vanessa P.G. Bretas <sup>a, b, c, \*</sup>, Ilan Alon <sup>a</sup>, Thelma Valéria Rocha <sup>b</sup>, Jefferson R. B. Galetti <sup>c</sup>

<sup>a</sup> School of Business and Law, University of Agder, Kristiansand, Norway

<sup>b</sup> PPGA, ESPM, São Paulo, Brazil

<sup>c</sup> DST/NRF South African Chair in Industrial Development, University of Johannesburg, Johannesburg, South Africa

#### ARTICLE INFO

Keywords: International franchising Governance modes Modes of entry International marketing International business Emerging markets

#### ABSTRACT

The purpose of this article is to test a novel, integrative theory of governance mode selection in the context of international franchising from Brazil. Given the Brazilian emerging market context, we added and tested another variable in the model relating to environmental distance between the home (Brazil) and host countries. To test the theory, we employed two logistic regression models and representative data from the Brazilian Franchising Association in addition to the World Bank's Doing Business index, and the CEPII (Centre d'études prospectives et d'informations internationales). Findings show that three factors influence Brazilian franchisors' choices regarding international governance modes: the environmental uncertainty of the host countries, their behavioral uncertainty, and the franchisors' financial capacity. However, the environmental distance does not influence internationalization strategies of Brazilian firms. Emerging markets firms' governance modes of entry can be adequately explained with the existing model despite contextual differences.

#### 1. Introduction

One of the most discussed areas in the franchising literature is international governance (Rosado-Serrano et al., 2018). When a company adopts franchising as a way of entering foreign markets, it can choose different types of governance modes that vary in terms of residual income, ownership, and the allocation of decision rights (Jell-Ojobor and Windsperger, 2014). Due to the complexity of these decisions, Jell-Ojobor and Windsperger (2014) proposed an integrative multi-theoretical model to assess both the organizational economics and strategic management factors that might influence the franchisors' choice. They evaluated the model empirically through a case study with two franchisors from Germany and France (Jell-Ojobor and Windsperger, 2017), but the authors suggested conducting quantitative research in different markets.

Following this suggestion, this study quantitatively tests the conceptual model in an emerging market. Brazil offers an interesting context in which to investigate the franchising governance mode choices, both empirically and theoretically. Empirically it is interesting because most of the work on franchising over the last two decades has used data from the United States, Canada, and Western Europe, despite the relevance and growth potential of franchisors headquartered in emerging economies (Aguiar et al., 2017; Lafontaine, 2014; Nielsen et al., 2018; Wright et al., 2005). Few studies of international franchising from Brazil exist, but the market is

\* Corresponding author.

E-mail address: vanessab@uia.no (V.P.G. Bretas).

https://doi.org/10.1016/j.intman.2021.100851

Received 2 April 2020; Received in revised form 11 February 2021; Accepted 30 March 2021

Available online 15 April 2021

<sup>1075-4253/© 2021</sup> The Author(s). Published by Elsevier Inc. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

large and growing. In 2019, Brazil had 2918 franchisor brands with 160,958 operations, and revenue growth of 6.8%. Moreover, from 2010 to 2019, the number of Brazilian franchisors in international markets grew by 151% (ABF, 2020).

From a theoretical perspective, internationalizing Brazilian franchisors represent emerging market multinationals, which possess unique characteristics in their international expansion strategies contributing to the Goldilocks' debate about the suitability of existing theories (Cuervo-Cazurra, 2012). Being used to unstable conditions and weak institutions may help emerging market multinationals in similar environments, such as those encountered in other developing countries (Cuervo-Cazurra, 2012; Cuervo-Cazurra and Genc, 2008; Ramamurti, 2012). Home country conditions such as an uncertain environment might prompt companies headquartered in emerging markets to follow particular paths of international expansion with greater internalization on other emerging market locations.

Our article contributes to the debate by siding with the camp, argued by Ramamurti (2009), suggesting that emerging markets multinationals follow a particular path of internationalization that can be partly explained by predictions of existing theories. Emerging economies are diverse, with different institutional arrangements and heterogeneous firms. The environmental similarity may not be the only or primary aspect influencing the internationalization decisions of all emerging market companies. This is true of Brazilian franchisors. Other factors, as the host country business environment, may become more relevant to the companies' expansion strategies (Estrin et al., 2018; Hoskisson et al., 2013; Lanfranchi et al., 2020; Melo et al., 2015; Wright et al., 2005).

For instance, the most important destination country for Brazilian franchisors is the United States because of its market potential and business efficiency (Cantoni et al., 2019; Melo et al., 2015). In 2019, of 163 international Brazilian franchisors, 67 had operations in the United States (ABF, 2020). In Aguiar et al.'s (2014) study, 7 of the 21 Brazilian franchisors interviewed had the US as their first foreign host market. Brazilian companies regard the US as their reference country (Fonseca and Rocha, 2018; Melo et al., 2015), despite the differences in their business environments. We conclude that the internationalization models need extensions or refinements only when the aspects of the country-of-origin are relevant (Cuervo-Cazurra, 2012).

This article aims to understand to what extent the integrative model of the international governance mode can explain the decisions of Brazilian franchisors. To achieve this goal, we tested Jell-Ojobor and Windsperger's (2014) conceptual model empirically using data from Brazil. We expand the boundaries of the model by testing the impact of the environmental distance between the home and host countries on the choice of governance mode of Brazilian franchisors. We conducted our quantitative analysis with a logistic regression model using Brazilian Franchising Association data from 108 franchisors with international operations, the World Bank's Doing Business index, and the CEPII (Centre d'études prospectives et d'informations internationales) databases.

This study contributes to the international franchising and emerging markets literature in three ways. First, we investigated the role of the environmental distance between the home and host countries in the choice that Brazilian franchisors made about their governance mode to determine whether the integrative model of the choice of international franchising governance mode needs to be extended to the emerging market context. Furthermore, we tested the integrative model through quantitative research, providing statistical evidence to complement the theoretical and partial results from analyses of case studies provided by previous research (Jell-Ojobor and Windsperger, 2017, 2014). Finally, most research on emerging markets multinationals is in the context of industrial companies and in particular Chinese ones (Alon et al., 2018). This study concerns service companies whose products are often unexportable because consumption and production are inseparable and, thus, home country endowments will play a lesser role.

The rest of this article is organized as follows. After this introduction, the next section presents the theoretical background. Section 3 introduces the model and hypotheses. Section 4 outlines the methodology adopted. Section 5 describes the empirical test of the hypotheses and results. Section 6 discusses the findings, and Section 7 presents the conclusions of the study.

#### 2. Theoretical background

#### 2.1. International entry and governance modes

The literature about international business, market entry, and retailing classifies franchising as an important mechanism through which companies enter international markets (Cavusgil et al., 2017; Jell-Ojobor and Windsperger, 2014; Mumdžiev, 2011). According to the hierarchical model of market entry modes (Pan and Tse, 2000), companies must make two decisions about how to enter new markets. First, they must choose between equity and non-equity modes. Second, they must choose the specific mode of entry. Possibilities include joint ventures, wholly-owned subsidiaries, exports, and contractual agreements. Franchising is classified as a non-equity, contractual, medium-control entry mode (Brown et al., 2003; Kedia et al., 1994; Sashi and Karuppur, 2002).

A company that adopts franchising as a way to enter foreign markets can develop its operations abroad through various governance modes, meaning the organizational arrangements that companies utilize to conduct their international business (Benito et al., 2009; Jell-Ojobor and Windsperger, 2014). The main international franchising governance modes are direct investment, joint ventures of franchising, direct franchising, master franchising, and area development. These modes differ in terms of the allocations of residual income and control rights (ownership and decision rights) between the franchisor and the partners in foreign countries. The degree of the franchisor's investment required and the level of control increase from modes such as master franchising and area development to joint ventures of franchising and company-owned outlets because more residual income, ownership, and decision rights are allocated to the franchisor. Therefore, we can categorize the governance modes as high control or low control modes, based on the degree of control exercised by the franchisor and by its partners (Alon, 2000; Combs et al., 2011; Hansmann, 1996; Jell-Ojobor and Windsperger, 2014; Mumdžiev, 2011; Rosado-Serrano et al., 2018; Windsperger et al., 2009).

Direct investment (company-owned outlets) and joint ventures of franchising are examples of high control governance modes. In direct investments, the franchisor invests in company-owned outlets in foreign markets, maintaining full control over the operations.

In franchising joint ventures, the franchisor enters into an equity relationship with the foreign partner to set a joint venture company, making a master franchising or area development agreement with the joint venture company to expand the system. The franchisor and the joint venture partner share the ownership, decision, and residual income rights (Jell-Ojobor and Windsperger, 2014; Rosado-Serrano et al., 2018).

The major low control franchising governance modes are franchisee-operated arrangements. For instance, in direct franchising, the international franchisor sells the business concept to an individual franchisee in the host country. In a master franchising agreement, the franchisor grants to a master franchisee the right to operate franchise units and to give franchise rights to third parties. In an area development agreement, the franchisor grants the franchisees a defined territory in which they can develop units. In these modes, the local partner has only decision and residual income rights (Jell-Ojobor and Windsperger, 2014; Rosado-Serrano et al., 2018).

Several previous studies have dealt with governance modes and the factors that influence their choice using different theoretical lenses (Aguiar et al., 2017; Alon, 2000; Benito et al., 2009; Burton et al., 2000; Dikova and Brouthers, 2016; Jell-Ojobor and Windsperger, 2017, 2014; Mumdžiev, 2011). However, the acknowledged relevance of theoretical diversity in explaining complex phenomena such as international entry and governance modes (Anil et al., 2014; Brouthers and Hennart, 2007; Chiao et al., 2010; Combs et al., 2004; Dikova and Brouthers, 2016; Gaur et al., 2014; Merrilees, 2014; Rosado-Serrano et al., 2018) is evident in few multi-theoretical studies about international franchising governance modes, and most of these articles are conceptual, descriptive, exploratory studies (Aguiar et al., 2017; Alon, 2000, 2006a; Jell-Ojobor and Alon, 2017; Jell-Ojobor and Windsperger, 2014, 2017; Merrilees, 2014).

Moreover, the majority of the entry and governance mode studies have been focused on developed countries, despite the growing relevance of emerging markets, and manufacturing, despite the growing importance of the service sector. Franchise chains from emerging markets have different historical, cultural, economic, and social contexts. Their particular institutional environment requires further investigation to better understand their paths of international expansion (Aguiar et al., 2017; Lafontaine, 2014; Melo et al., 2019; Nielsen et al., 2018).

#### 2.2. Specificities of emerging markets

The increasing academic interest in multinationals headquartered in developing and emerging markets has generated a debate on whether the existing theories based on the analysis of such firms in advanced economies can explain these companies' strategies or whether they require new and specialized theories. One view argues that existing theories can explain both developed and emerging market firms' decisions (Rugman, 2010). A second view asserts that emerging market firms behave differently, and new theories are needed (Guillén and García-Canal, 2009; Mathews, 2006). A third view claims that multinationals based in developing and emerging countries can help modify and extend the existing theories (Alon et al., 2011; Cuervo-Cazurra, 2012; Cuervo-Cazurra and Genc, 2008; Cuervo-Cazurra and Montoya, 2018; Ramamurti, 2012).

Regarding the international franchising governance mode choice, the Goldilocks debate engenders the need to investigate if the mainstream theoretical explanations derived from advanced economies are adequate to explain the decisions of emerging market franchisors. The franchisor's governance mode decisions are intrinsically related to the institutional environment of the host markets. The traditional international franchising literature posits that host markets' environmental uncertainty increases the transaction costs associated with monitoring and coordination activities. Therefore, franchisors allocate more ownership and decision rights to local partners to count on their knowledge and experience, choosing high control modes in underdeveloped and weak institutional environments (Hoffman et al., 2016; Jell-Ojobor and Windsperger, 2017, 2014).

However, emerging markets have environmental uncertainties borne out of lesser economic development and weaker institutional settings. Malfunctioning or missing institutions prevalent in emerging economies increase transaction costs. The institutional voids (where existing institutional arrangements and support are absent, weak, or fail) require the development of firm capabilities to address and manage these voids and their related transaction costs. Emerging markets multinationals, as a result, are required to be flexible, willing to take risks, and develop a higher tolerance and resilience to instability (Cuervo-Cazurra, 2012; Doh et al., 2017; Khanna et al., 2005; North, 1991, 1990; Ramamurti, 2012).

The ability to interact with institutional voids in their domestic market may endow emerging market franchisors with greater competencies in dealing with transaction costs as compared to firms from developed economies. The similarity between the emerging markets' institutional environments may help their firms to deal with authoritarian political regimes, corruption, fluctuation in law enforcement, and other related issues (Aguiar et al., 2017; Cuervo-Cazurra, 2012; Cuervo-Cazurra and Genc, 2008; Lanfranchi et al., 2020; Melo et al., 2019).

Thus, the conditions in the emerging markets' home country would affect the ability of their companies to deal with the transaction costs involved in similar environments and influence decisions about host market selection and governance modes. For this reason, emerging market companies would prioritize host countries with similar institutional environments and internalize foreign operations more efficiently, adopting high control modes in other emerging markets.

Nevertheless, despite having typical features such as the experience dealing with uncertainties in their home country, emerging markets are heterogeneous and differ in initial conditions and competitive outcomes. Since they have different levels of institutional and factor market development, the transaction costs and resources needed to achieve competitive advantage vary significantly. The heterogeneity implies that transaction costs also exist between different emerging countries (Estrin et al., 2018; Hoskisson et al., 2013; Kedia et al., 2015; Wright et al., 2005).

Besides, Brazil is a mid-range emerging economy, with institutions, infrastructure, and factor market semi-developed. Firms from mid-range emerging economies consist of a diverse group of companies with significant organizational heterogeneity, and for this

reason, they follow different strategic options (Hoskisson et al., 2000, 2013; Wright et al., 2005). As a result, some Brazilian companies seek host countries with similar environments, and others might prioritize better market and institutional conditions in their expansion strategies. The similarity between home and host countries may not be the only factor influencing their governance mode choices. Other factors can play a role in their decisions, as the market potential and the host country environment, despite the differences in their business environments.

Thus, not all behaviors of Brazilian franchisors are necessarily different from those of companies in advanced economies. They may share many characteristics. For this reason, we investigate the Brazilian franchisors' international governance mode decisions in light of multidisciplinary general theories derived from advanced economy studies, including features about the specificities of the emerging markets.

#### 3. Integrative model and hypotheses

Jell-Ojobor and Windsperger (2014) proposed an integrative model that includes organizational economics and the strategic management variables that can influence the choice of international governance mode. The organizational economics perspective highlights the effect of the choice of governance mode in reducing costs and potentially increasing residual income. The strategic management perspective focuses on the impact of this choice on the company's competitive advantage. Thus, the integrative model considers both concerns related to reducing costs and the issue of competitive advantage.

From the organizational economics perspective (transaction cost and agency theories), the authors analyze environmental uncertainty, behavioral uncertainty, and transaction-specific investments. The strategic management perspective (resource-based, organizational capability, and property rights theories) considers system-specific assets, local market assets, financial assets, and intangible assets. Jell-Ojobor and Windsperger (2017) empirically evaluated the integrative model through a case study analysis of the European automotive rental industry, and Hoffman et al. (2020) tested the model in a sample of US franchisors. The findings showed that both organizational economics and strategic management factors are relevant for developed country franchisors' choice of governance mode.

#### 3.1. Transaction cost theory and agency theory perspectives on the choice of international franchise governance mode

According to transaction cost theory and agency theory, the choice of governance mode in international markets is related to the franchisors' goal of minimizing costs. Several transaction and agency costs can be associated with international franchising. Examples include the costs of monitoring international operations, transferring knowledge to distant markets, and protecting property rights, as well as possible opportunism or shirking by the local partners. Thus, franchisors choose the governance mode most likely to reduce these costs, given factors such as the environmental uncertainty of the host markets and the behavioral uncertainty of the franchisees (Baena, 2012; Burton et al., 2000; Contractor and Kundu, 1998a; Fladmoe-Lindquist and Jacque, 1995; Grewal et al., 2011; Jell-Ojobor and Windsperger, 2014; Mumdžiev, 2011).

In international markets with different socio-cultural, economic, and institutional environments, the costs related to monitoring, adaptation, information transfer, and property rights protection are higher. The uncertainties and risks of host markets make it difficult to control and coordinate foreign operations. The franchisors' lack of information about the specificities of host countries that suffer from environmental uncertainty requires them to delegate more tasks to their local partners (Hoffman et al., 2016; Jell-Ojobor and Windsperger, 2017, 2014).

In other words, the environmental uncertainty of host markets increases the franchisors' transaction costs related to monitoring and coordination activities. Therefore, they will be inclined to allocate more decision rights to franchisees. As a result, in host countries with greater environmental uncertainty, franchisors that choose lower control modes count on their local partners' knowledge about the specificities of those markets and give them the flexibility to deal with changes (Baena, 2012; Burton et al., 2000; Fladmoe-Lindquist and Jacque, 1995; Jell-Ojobor and Windsperger, 2014; Mumdžiev, 2011). Thus, the transaction costs incurred due to environmental uncertainty in host markets increase the likelihood that franchisors will choose lower control modes (Jell-Ojobor and Windsperger, 2017, 2014; Mumdžiev, 2011). Hence, we posit the following hypothesis:

## **Hypothesis 1a**. The greater the environmental uncertainty of the host country, the more likely a franchisor will choose a low control mode in foreign markets.

We extend the model of Jell-Ojobor and Windsperger (2014) by including features about the specificities of the emerging markets, which might be relevant for internalization or modes of entry. For some scholars, being used to institutional voids can enhance the emerging market firms' ability to deal with transaction costs in similar environments (Cuervo-Cazurra, 2012; Ramamurti, 2012). Under this condition, emerging markets multinationals might internalize foreign operations in other emerging markets that have similar institutional arrangements. In other words, franchisors will choose high control modes since they have experience on operating in environments that are similar to their home country economy, depending less on local partners. Conversely, when the environmental distance between home and host countries increases, franchisors are inclined to allocate more decision rights to local franchises to count on their knowledge and experience.

Thus, we propose a second model (Model 2) that includes the environmental distance between the home and host country instead of the environmental uncertainty variable of Model 1, according to following hypothesis:

Hypothesis 1b. The greater the environmental distance between the home and host countries, the more likely the Brazilian franchisors choose

#### a low control mode in foreign markets.

Successfully transferring information to distant locations is a challenge to franchisors, increasing the level of behavioral uncertainty. When the geographic distance between the home and host countries is greater, the information asymmetry and the risks of opportunistic behavior, free-riding, and shirking are greater (Castrogiovanni et al., 2006; Combs et al., 2004; Fladmoe-Lindquist and Jacque, 1995; Jell-Ojobor and Windsperger, 2017, 2014; Mumdžiev, 2011; Watson et al., 2005).

Hence, the greater the geographic distance between home and host country, the greater the behavioral uncertainty and the agency problems derived from it. The geographic distance increases the monitoring costs needed to reduce opportunistic behavior and the agency costs related to the selection of poor employees abroad and possible moral hazards. By choosing lower control modes, franchisors shift the costs of these problems to the local partner. The franchise contracts align the incentives of the local partner with those of the franchisor and reduce agency costs (Contractor and Kundu, 1998a; Fladmoe-Lindquist and Jacque, 1995; Grewal et al., 2011). Thus, lower control modes can reduce monitoring costs and mitigate agency problems:

**Hypothesis 2**. The greater the behavioral uncertainty of the host country, the more likely a franchisor will choose a low control mode in foreign markets.

#### 3.2. Resource-based theory and organizational capability theory perspectives on the choice of international franchise governance mode

Resource-based theory and organizational capability theory perspectives on the choice of international franchise governance modes focus on the creation of value from the franchisors' assets. Franchisors gain a competitive advantage by exploring and exploiting firm-specific resources and capabilities (Erramilli et al., 2002; Hussain et al., 2018). When a firm enters a foreign market, it must transfer its resources and capabilities to foreign operations. Therefore, it must choose an entry and governance mode that can accomplish these goals effectively (Erramilli et al., 2002; Hussain et al., 2018; Jain et al., 2015; Jell-Ojobor and Windsperger, 2017, 2014; Moraes et al., 2008).

The resource-based perspective point outs that franchisors must have sufficient resources such as financial capital, brand name recognition, and managerial and routine-processing know-how before internationalizing (Mariz-Pérez and García-Álvarez, 2009; Perrigot et al., 2004). Therefore, the allocation of these critical resources and capabilities needed to operate in foreign markets, such as system-specific and financial assets, between the franchisor and partners in international markets influences the choice of governance mode (Jell-Ojobor and Windsperger, 2014).

Financial assets are related to the ability of the franchisor and the local partners to fund the operations in the host market. Franchisors that need capital to grow use franchising to obtain resources for their expansion. Therefore, they will rely on investment fees and royalties to fund their expansion. If the franchisor has financial constraints, the local partner's resources will be necessary for the international expansion. The international expansion increases the franchisor's costs significantly (Castrogiovanni et al., 2006; Combs et al., 2004; Jell-Ojobor and Windsperger, 2017, 2014).

Lower control modes such as master franchising and area development agreements can make international expansion less expensive for the franchisor. Thus, we posit that franchisors with limited financial resources to internationalize their operations will favor lower control modes and count on their local partners' financial assets. On the other hand, franchisors that have financial resources from their domestic operations might choose high control modes in their international operations (Combs et al., 2004; Jell-Ojobor and Windsperger, 2017, 2014; Melo et al., 2015).

## **Hypothesis 3.** The greater the franchisor's financial ability to invest in international expansion, the greater the likelihood of choosing a high control mode in foreign markets.

System-specific assets are related to the franchisor's resources and capabilities. Other factors important in this area are capabilities that are difficult to codify, transfer, and imitate such as business know-how, brand name, marketing, R&D capabilities, advertising, and promotion. When the franchisor's system-specific assets are more relevant than the franchisee's local market assets in creating residual income, the franchisor will retain the decision rights (Castrogiovanni et al., 2006; Fladmoe-Lindquist, 1996; Jell-Ojobor and Windsperger, 2017, 2014; Windsperger and Dant, 2006).

One of the essential system-specific assets of the franchisor is brand name. Previous studies have underscored the importance of intangible resources such as brand name and reputation for the success of a franchise network. The research highlights brand reputation as a franchisor-owned strategic asset that gives franchisors a competitive advantage in international expansion (Barthélemy, 2008; Elango, 2007; Fladmoe-Lindquist, 1996; Ghantous and Das, 2018; Gillis et al., 2014; Grewal et al., 2011; Perrigot et al., 2013; Rocha et al., 2018; Shane, 1996). Franchisors that count on a reliable brand name as an essential asset and the competitive advantage derived from it will be more likely to retain a higher level of control (Alon, 2006a, 2006b; Jain et al., 2015). Therefore, franchisors with a prestigious brand name are more likely to keep their decision and ownership rights.

Another system-specific asset critical to the franchising system is business know-how, including factors such as store layout, product mix, operating procedures, location selection priorities, monitoring capabilities, advertising, and R&D. Such know-how is unique to the franchisor, and thus considered a proprietary advantage. As with brand name, business know-how is difficult to transfer across firm boundaries and cannot be easily imitated. Therefore, it is quite likely to generate rents (Jell-Ojobor and Windsperger, 2017, 2014; Shane, 1996). Thus, we propose the following hypothesis:

**Hypothesis 4**. The more relevant the system-specific assets of the franchisor are, the greater the likelihood of choosing a higher control mode in foreign markets.

#### 4. Methodology

#### 4.1. Sample

To test our hypotheses, we used a quantitative, descriptive approach with a representative sample of data from the Brazilian Franchising Association – ABF – from 2018. According to ABF, in 2018, there were 145 Brazilian franchisors with international operations in 114 countries. The database used in this study provides information about 108 Brazilian franchisors, 74.5% of the total number of Brazilian franchisors operating abroad, and the 55 countries where those franchisors had operations in 2018. Thus, in total there are 267 observations in our study.

In addition, we also used the 2018 data from the World Bank's database Doing Business, which contains information regarding the institutional environment in the countries in which Brazilian franchisors have operations. We also utilized a database from CEPII (Centre d'études prospectives et d'informations internationales) with data about the geographic distance between home and host countries. We adopted the country classifications of the World Economic Outlook from the International Monetary Fund as to whether they were advanced economies or emerging and developing economies. All of our measures refer to 2018.

#### 4.2. Variables and measures

Table 1 summarizes the variables adopted, measures, and sources of data. The dependent variable is a dummy indicator of high control and low control modes in each foreign market. The variable received the value of 1 if the franchisor adopted at least one high control mode–company-owned outlets. Those that adopted only low control modes in each market–franchisee-owned outlets, master franchisee outlets, and area developers' outlets—received the value of 0.

The hypotheses involve several independent variables. *Environmental uncertainty* is a composite measure obtained from the World Bank's Doing Business database (Ang and Michailova, 2007; Dikova and Van Witteloostuijn, 2007; Michailova and Ang, 2008). As the business environment is a crucial factor in international franchising (Baena, 2012; Fladmoe-Lindquist and Jacque, 1995; Grewal et al., 2011), information from this source was suitable for assessing environmental uncertainty.

To compute the proxy for environmental uncertainty, we adapted to the franchising context previous composite measures adopted by entry mode studies to reflect the institutional environment (Ang and Michailova, 2007; Dikova and Van Witteloostuijn, 2007; Michailova and Ang, 2008), selecting specific indicators from the World Bank's database (ease of starting a business, protecting minority investors, enforcing contracts, and access to credit) used as driving factors influencing international franchising governance mode decisions (Baena, 2012, 2013; Contractor and Kundu, 1998a; Grewal et al., 2011; Lanfranchi et al., 2020).

Institutions and regulations related to the application of the law and contract enforcement are important as they protect the franchise chains' intangible assets, such as brand and know-how. Previous evidence shows the relationship between contract enforcement efficiency and international expansion via franchising (Baena, 2012, 2013; Contractor and Kundu, 1998a). An environment condition that facilitates business transactions, such as friendly credit and bureaucratic procedures, and good conditions to

#### Table 1

Variables, measures an	id sources of data
------------------------	--------------------

Variables	Measure	Sources
Dependent variable High-control modes (1)/low- control modes (0)	Dummy indicator: Franchisors that adopt at least one high-control mode in each market (1)/franchisors that adopt only low-control modes in each market (0)	Brazilian Franchising Association database (2018)
Independent variables Environmental uncertainty	Composite measure of the indicators: "ease of starting a business", "getting credit", "protecting minority investors", and "enforcing contracts"	World Bank's Doing Business index (2018)
Behavioral uncertainty	Geographic distance between home and host countries	CEPII (2011)
System-specific assets: Brand	Dummy indicator of the existence of an advertisement fee applied to franchisees, 0 or 1	Brazilian Franchising Association database (2018)
System-specific assets: Business know-how	Number of training courses and technical support areas	Brazilian Franchising Association database (2018)
Financial capacity	Composite measure of initial fee and collection of royalties	Brazilian Franchising Association database (2018)
Distance between home and host countries	Composite measure ("ease of starting a business", "getting credit", "protecting minority investors", and "enforcing contracts") of the distance between the 55 host countries and Brazil	World Bank's Doing Business index (2018)
Control variables		
Age	Franchisors' years of existence since their founding	Brazilian Franchising Association database (2018)
Network size	Number of outlets in the domestic market	Brazilian Franchising Association database (2018)
Sector	ABF classification into 11 sectors	Brazilian Franchising Association database (2018)
Host country classification	Dummy indicator of the country classification between "Emerging and developing economies" (1) and "Advanced economies" (0)	World Economic Outlook - IMF

small investors, is also crucial to franchise operations (Grewal et al., 2011; Lanfranchi et al., 2020).

We averaged each country's rank position for the selected indicators, and the resulting composite measure provides information about the country's performance relative to the performance of other countries in a way that higher scores represent greater environmental uncertainty ("Doing Business", 2019).

We measured *behavioral uncertainty* using the geographic distance between the home and host countries (Fladmoe-Lindquist and Jacque, 1995; Jell-Ojobor and Windsperger, 2017, 2014), based on the geographic definition of the "distance between two points on the surface of the earth, as given by latitudinal and longitudinal coordinates" (Beugelsdijk et al., 2018). To obtain the information needed for these calculations, we used a database from CEPII (Centre d'études prospectives et d'informations internationales) (Mayer and Zignago, 2011). We then calculated the geographic distance in kilometers between the 55 countries' capitals and Brazil's capital (Brasília).

*System-specific assets* consisted of the franchisors' brand name and business know-how. The proxy used to measure brand name is the existence of an advertisement fee applied to franchisees. Many empirical studies that evaluate the relevance of the brand to the international franchising from a resource-based perspective have used this proxy. The more important the franchisor's brand name assets, more revenues from advertising fees are required to maintain the brand value through investments in communication and advertising (Lafontaine and Shaw, 2005; Mariz-Pérez and García-Álvarez, 2009; Perrigot et al., 2013; Windsperger, 2004; Windsperger and Yurdakul, 2007). The proxy used to measure the business know-how, also based on previous studies (Windsperger and Yurdakul, 2007), is the number of training courses and technical support areas that the franchisors offer to the franchisee. The number of training courses and technical support areas required is greater if the franchisor's business know-how is more specific and challenging to transfer.

The franchisees' investment, along with the on-going taxes charged by the franchisors in the domestic market, contributes to the franchisor's *financial capacity* to fund its international expansion. The franchisors report the minimum and maximum initial investment required to the Brazilian Franchising Association, and we calculated the average initial investment. The royalties are an on-going payment made by the franchisee. As there are several ways to charge royalties such as a fixed monthly fee, a percentage of gross sales, or a percentage per transaction, we considered just the existence of regular on-going payments, regardless of the collection mode. Thus, we devised a composite measure to proxy for the franchisors' financial assets that considered the existence of both an initial investment and on-going payments.

To measure institutional distance, international business researchers have usually used data from the World Bank, the Heritage Foundation EFI, and ICRG scores (Beugelsdijk et al., 2018). In order to include the effects of the home market and measure the *distance between the home and host countries*, we used the environmental uncertainty composite measure previously calculated, based on the four business environment indicators (ease of starting a business, protecting minority investors, enforcing contracts, and access to credit) to compute the Euclidean distance between each of the 55 host countries and Brazil's position in the composite measure. For cultural and institutional distances, Euclidean distance results do not differ from other distance indexes (Beugelsdijk et al., 2018), and it is widely employed in international business studies (Barkema and Vermeulen, 1997; Drogendijk and Slangen, 2006). The smaller the absolute value of the distance between Brazil and a specific country, the more similar their business environment. For instance, as Brazil is a mid-range economy, it is nearer to other BRICS countries, such as India and South Africa, and more distant from both developed countries as Australia and poor ones as Guinea-Bissau.

In addition to these independent variables, we also controlled for four other variables. First, we considered the franchisors' age, calculated as the number of years since their founding. As firms get older, they gather more experience in partner and site selection, monitoring, and control, which is essential to develop international operations (Alon et al., 2012, 2013; Alon and McKee, 1999; Burton et al., 2000; Dikova and Brouthers, 2016; Madanoglu et al., 2017; Ni and Alon, 2010).

Second, we controlled for the size of the network, measured by the number of domestic operations of the Brazilian franchisors. Several studies have documented that a firm's size affects its choice of mode of entry and governance. The size indicates the availability of resources that a firm can commit to the international operations and also its ability to manage distant localized partners, influencing the governance mode decision (Alon and McKee, 1999; Brouthers and Hennart, 2007; Burton et al., 2000; Contractor and Kundu, 1998a, 1998b; Dikova and Brouthers, 2016; Madanoglu et al., 2017). Other studies have established that a firm's size affects the relationship between the institutional aspects of the host countries and the level of commitment abroad (Melo et al., 2019). We also controlled for the franchisors' sectors according to the ABF classification that categorizes business activities into 11 sectors: food; construction; technology and electronics; recreation; travel and hotel; cleaning and maintenance; fashion; health and personal care; automotive; business services; and education.

In Model 2, we controlled for the host countries' classification as an advanced economy or an emerging and developing economy by assigning a dummy variable the value of 1 if the company was from an emerging country. By measuring the *distance between the home and host countries* and controlling for the host countries' classification, we sought to determine whether the similarity between the home and host countries' environments affected the choice of international governance mode (Cuervo-Cazurra, 2012; Cuervo-Cazurra and Genc, 2008; Surdu et al., 2018; Wright et al., 2005).

#### 4.3. Research design

Table 2 shows the means, standard deviations, and correlation coefficients for all of the variables under study. We conducted a binomial logistic regression analysis in order to test the hypotheses presented. We used this statistical model because of the dichotomous nature of the dependent variable and the mix of continuous and categorical independent variables (Hair et al., 1995). We followed other studies that used logistic regressions to investigate choices about international entry and governance modes (Ang and

Table 2	
Means standard deviations an	d correlations

œ

wichnis, standard deviations and e	correlations.											
Variables	Mean	s.d.	1.	2.	3.	4.	5.	6.	7.	8.	9.	10.
1. High control mode	0.19	0.39	1									
2. Environmental _uncertainty	99.1	49.5	-0.33**	1								
3. Home and host distance	35.8	20.9	0.01	0.15*	1							
4. Behavioral uncertainty	5873	3657	0.05	-0.51**	0.00	1						
5. System-specific: Brand	0.75	0.44	0.00	0.03	0.08	0.10	1					
6. System-specific: Knowhow	27.1	6.9	-0.03	-0.06	-0.11	0.09	-0.01	1				
7. Financial capacity	324,731	253,302	0.04	0.13*	0.03	-0.14*	0.01	$-0.32^{**}$	1			
8. Host country _classification	0.61	0.49	-0.30**	0.70**	-0.06	-0.55**	0.03	-0.03	0.25**	1		
9. Size	378	702	-0.03*	0.16**	0.09	0.00	0.15*	-0.12	0.31**	0.15*	1	
10. Age	25.4	19.8	-0.14	0.21**	0.03	-0.18**	-0.01	-0.06	0.34**	0.16**	0.30**	1

Note: High control mode (1 = high-control), and 0 = low-control); System-specific: Brand (1 = apply advertisement fee, and <math>0 = do not apply); Financial: Royalties (1 = apply royalties, and <math>0 = do not apply); Host country classification (1 = emerging and developing economies, and <math>0 = advanced economies). Coefficients are statistically significant at the \*p < 0.05; \*\*p < 0.01 levels.

Michailova, 2007; Chiao et al., 2010; Contractor and Kundu, 1998a, 1998b; Dikova and Van Witteloostuijn, 2007; Erramilli et al., 2002; Fladmoe-Lindquist and Jacque, 1995; Meyer et al., 2009).

In the binomial logistic regression, the regression coefficients estimate the impact of the independent (or control) variables on the probability that the international franchishor will choose a high control mode (which carries the value of 1). The model can be expressed as

$$P(Y) = \frac{1}{1 + e^Z} P,$$
 (1)

where Y is the dependent variable, and Z is the linear combination of the independent and control variables. That is,

$$Z = \beta_0 + \beta_1 X_1 + \beta_2 X_2 + \dots + \beta_n X_n,$$
(2)

where  $\beta_0$  is the intercept,  $\beta_1 \dots \beta_n$  are the regression coefficients and  $X_1 \dots X_n$  are the independent and control variables. To facilitate the comparisons, we normalized all of the non-binary variables by subtracting the mean and dividing by the standard deviation.<sup>1</sup> Another econometric issue is that the data might be correlated with several operations of the same franchisor. To address this issue, we used robust standard errors clustered at the firm and country-of-operation level (Cameron and Trivedi, 2005).

We conducted the analysis by creating two models. Model 1 estimates the effects of the first set of predictor variables (environmental uncertainty, behavioral uncertainty, system-specific assets, and financial assets) on the likelihood of a firm choosing a high control mode. Model 2 estimates the effects of the second set of predictor variables (home and host countries' environmental distance, behavioral uncertainty, system-specific assets, and financial assets) on the likelihood of choosing a high control mode. By comparing the two models, we can determine whether the similarity between the home and host countries has an impact on the choice of international governance mode. If there is an effect, the result might suggest that firms from emerging and developing countries behave differently than those from advanced economies. In contrast, if there is no effect, the result might suggest that Brazilian franchisors follow the same patterns reported in the traditional governance mode literature.

#### 5. Results

Table 3 reports the results for Models 1 and 2.

#### 5.1. Hypotheses related to the theories of transaction and agency costs

First, we investigate which model fits well with the emerging markets franchisors' decisions on the governance modes. In Model 1, we tested the influence of the *host countries' environmental uncertainty* in governance mode choice. The results of the Model 1 support Hypothesis 1a. Consistent with the traditional international franchising governance mode literature based on transaction cost theory and agency theory (Baena, 2012; Fladmoe-Lindquist and Jacque, 1995; Jell-Ojobor and Windsperger, 2014), environmental uncertainty had a significant (p < 0.01) negative relationship with the likelihood of choosing a high control mode. Thus, when the environmental uncertainty increases, the probability of a firm choosing a high control mode decreases.

Moreover, considering all of the variables on average, the likelihood of an outstanding franchisor choosing a high control mode was 8.67%. The probability of choosing a high control mode increased to 29.0% if the franchisors operated in countries with less environmental uncertainty (all other variables remained constant). In other words, considering that higher scores in the composite measure correspond to greater environmental uncertainty, in countries with 1 standard deviation below the average franchisors are more likely to choose a high control mode. For instance, they are more likely to select a low control mode in India and South Africa than Ireland and France.

The results of Model 2 for the *distance between home and host countries* were not statistically significant. As a matter of robustness, we also tested the distance between the home and host countries measured with Kogut and Singh's index, but the results remain not statistically significant. Thus, Hypothesis 1b is not supported. Findings of Models 1 and 2 imply that Brazilian franchisors consider the environmental uncertainty of host countries a primary factor in their choice of the governance mode, regardless of the home country's conditions and its relative distance to the host countries. Thus, the traditional transaction cost theory and agency theory arguments for the choice of international governance mode choice also apply to the franchisors from Brazil that internationalize their operations. Similarities between the home and host countries are not relevant to Brazilian franchisors' decisions about their international governance mode.

Moreover, the control for the host countries' classification was also significant in Model 2 (p < 0.01) and indicated a negative relationship between operating in emerging and developing host countries and the likelihood of choosing a high control mode. In other words, Brazilian franchisors do not choose high control modes in other emerging and developing countries because of their similar home country conditions. With all variables on average, the probability of choosing high control declined from 37.0% when the host country was an advanced economy to 4.13% when the host country was an emerging or developing economy.

<sup>&</sup>lt;sup>1</sup> Transforming this variable also helps avoid the potential multicollinearity issue that could appear, especially when including the additional interaction terms. Indeed, the variance inflation factor (VIF) used to identify the presence of multicollinearity between variables is below the suggested threshold value of 10 (Baum, 2006).

#### Table 3

Logistic regression results.

Dependent variable: high control mode				
Variables	Model 1	Model 2		
Environmental uncertainty	-1.45970***			
	(0.30116)			
Home and host distance		-0.07095		
		(0.30325)		
Behavioral uncertainty	-0.76341**	-0.69570**		
	(0.35236)	(0.29900)		
System-specific: brand	1.14492	1.17016*		
	(0.75228)	(0.66901)		
System-specific: know-how	0.27308	0.33629		
	(0.33846)	(0.31944)		
Financial capacity	0.50072*	0.78083**		
	(0.29003)	(0.30367)		
Host country classification		$-2.61340^{***}$		
		(0.51589)		
Size	0.46286*	0.48076**		
	(0.25401)	(0.23556)		
Age	$-1.39836^{***}$	$-1.60067^{***}$		
	(0.54140)	(0.56213)		
Constant	-2.35410***	-0.53188		
	(0.80655)	(0.98438)		
Observations	267	267		
Sector fixed effects	Yes	Yes		
Log Likelihood	-84.48289	-86.29970		
Overall Chi-square	91.459***	87.825***		
Nagelkerke R <sup>2</sup>	0.4656	0.4500		

Note: The dependent variable high control mode equals 1 if firm *i* chooses to enter the host country market with a high control mode, and zero otherwise. All of the non-binary independent variables are standardized by subtracting the mean and dividing by the standard deviation. Heteroskedasticity-robust standard errors (clustered at the firm level and operationalized by the country level) are shown in parentheses. Coefficients are statistically significant at the \*p < 0.1; \*\*p < 0.05; \*\*\*p < 0.01 levels.

Behavioral uncertainty, measured by the geographic distance, had a significant (p < 0.05) and negative relationship with the likelihood of choosing a high control mode. Thus, in countries that are further from the home country, franchisors are more likely to choose a low control mode. As the geographic distance from the home market increases, so does behavioral uncertainty. This result supports Hypothesis 2. The probability of choosing a high control mode increased from 8.67%, with all variables on average, to 16.93% when Brazilian franchisors operated in countries closer to home (1 standard deviation less than the average distance and all other variables remain constant). For example, they are more likely to select a low control mode in Guatemala and Angola than Uruguay and Argentina.

#### 5.2. Hypotheses related to the resource-based and organizational capabilities theories

The significant (p < 0.1) and positive relationship between the companies' financial capacity and the likelihood of choosing a high control mode provide support for Hypothesis 3. Given their revenues in the domestic market, franchisors have more financial capacity to fund their international expansion, increasing the probability of a higher control mode. Thus, our results support the literature claiming that franchisors with limited financial resources to internationalize their operations will favor lower control modes and count on their local partners' financial assets (Combs et al., 2004; Jell-Ojobor and Windsperger, 2014, 2017).

Regarding Hypothesis 4, the results for brand name and business know-how were not statistically significant, although their coefficients were positive, as expected. Thus, these results do not support the hypothesis that system-specific assets, specifically, brand name and business know-how, increase the probability of choosing a high control mode in foreign markets.

The results of the franchisors' size as a control variable revealed a significant (p < 0.1) and positive relationship with the likelihood of a high control mode, meaning that franchisors with more units in the domestic market favor high control modes abroad. The firm's age had a negative and significant (p < 0.01) relationship with the likelihood of choosing a high control mode. In other words, franchisors with more years in business are more likely to adopt low control modes with local partners. Previous studies found mixed results regarding the effect of the franchisors' experience on the modes of entry and governance. Some reported that franchisors with more experience should have the resources to choose high control modes and avoid intermediary-related costs (Contractor and Kundu, 1998a, 1998b). Others did not find significant relationships (Erramilli et al., 2002). Our findings are aligned with Burton et al. (2000), who reported that more experienced franchisors employed intermediaries in their governance modes, and with Fladmoe-Lindquist and Jacque (1995), who documented a relationship between more experienced franchisors and the likelihood of using franchising contracts. Franchisors' experience and improved ability to select and monitor competent intermediaries despite the agency costs might explain the results.

#### 5.3. Robustness tests

To conduct robustness tests of Model 1, we used other variables related to the host country's environmental uncertainty as an independent variable. The results for Models 3 and 4 appear in Table 4. In the first test, we changed the composite measure of the business environment by using the 2019 OECD Country Risk Classification, which measures the country's credit risk and the likelihood that a country will service its external debt (Model 3). In the second test in Model 4, we changed the composite measure of the business environment by using the GDP per capita based on purchasing power parity (constant in 2011 international dollars). Several papers have used country risk and economic development (GDP) as proxies for environmental uncertainty (Alon et al., 2010; Brouthers and Hennart, 2007; Contractor and Kundu, 1998a, 1998b; Morschett et al., 2010). Given that both of these factors are associated with measures of the business environment, we substituted them in Model 1 to check if the effects remained the same.

Results for Model 3 showed that country risk had a significant (p < 0.01) and negative relationship with the likelihood of choosing a high control mode. The riskier the host country, the greater the probability of Brazilian franchisors choosing low control modes. Furthermore, the results for Model 4 indicated that the lower the GDP per capita of the host countries, the greater the probability of low control modes. These results are in line with our previous results.

Our findings support the behavioral uncertainty and financial capacity hypotheses in all models. Results for business know-how remained not statistically significant. However, in Model 3, the results for brand name were positive and statistically significant, consistent with Hypothesis 4.

#### 5.4. Extensions

In this section, we extend the analysis based on our earlier findings. Previous results demonstrated that the environmental uncertainty of host countries has an impact on the Brazilian franchisors' choice of international governance mode but the distance between the home and host countries does not influence their decision. Thus, considering Model 1 as our basic model, we modified it to include the interactive effects between environmental uncertainty and the impact of brand name (Model 1.1), business know-how (Model 1.2), financial capacity (Model 1.3), and behavioral uncertainty (Model 1.4) on the choice of governance mode.

Jell-Ojobor and Windsperger (2014, 2017) suggested negative interactive effects between environmental uncertainty and the impact of financial assets and behavioral uncertainty on the franchisor's tendency to use high control modes. Furthermore, our previous results showed that the impact of system-specific assets such as brand name and business know-how have no significant impact on the choice of international governance modes of Brazilian franchisors. As some scholars argue, the impact of a firm's specific resources on this choice is not uniform, and the firm's advantages may vary depending on the country. Institutional dimensions might

#### Table 4

Logistic regression results - robustness tests.

Dependent variable: high control mode					
	Model 1	Model 3	Model 4		
Environmental uncertainty	-1.45970***				
	(0.30116)				
Risk		-1.40880***			
		(0.29539)			
GDP per capita			1.09179***		
			(0.20183)		
Behavioral uncertainty	-0.76341**	-0.70342**	-0.60767**		
	(0.35236)	(0.30544)	(0.29100)		
System-specific: Brand	1.14492	1.29769*	0.88266		
	(0.75228)	(0.68994)	(0.71152)		
System-specific: Know-how	0.27308	0.27266	0.24849		
	(0.33846)	(0.32306)	(0.33625)		
Financial capacity	0.50072*	0.64053**	0.65053**		
	(0.29003)	(0.32377)	(0.28096)		
Size	0.46286*	0.51659**	0.44404*		
	(0.25401)	(0.25569)	(0.23741)		
Age	-1.39836***	-1.50472***	-1.68677**		
	(0.54140)	(0.56975)	(0.65765)		
Constant	-2.35410***	-2.33321***	-1.97451**		
	(0.80655)	(0.86484)	(0.77527)		
Observations	267	267	267		
Sector fixed effects	Yes	Yes	Yes		
Log likelihood	-84.48289	-85.83874	-86.79767		
Overall Chi-square	91.459***	88.747***	86.829***		
Nagelkerke R <sup>2</sup>	0.4656	0.4540	0.4456		

Note: The dependent variable high control mode equals 1 if firm *i* chooses to enter the host country market with a high control mode, and zero otherwise. All of the non-binary independent variables are standardized by subtracting the mean and dividing by the standard deviation. Heteroskedasticity-robust standard errors (clustered at the firm level and operationalized by the country level) are shown in parentheses. Coefficients are statistically significant at the \*p < 0.1; \*\*p < 0.05; \*\*\*p < 0.01 levels.

also influence the ability of companies to exploit specific resource-based advantages (Brouthers et al., 2008; Hennart and Slangen, 2015). For this reason, we also tested the interactive effects between environmental uncertainty and the impact of brand name and business know-how on the franchisor's tendency to use high control modes.

The results presented in Table 5 revealed two significant interactive effects.

Interactive effects between environmental uncertainty and business know-how. The environmental uncertainty of host countries had a significant (p < 0.1) interactive effect on the relationship between the business know-how of Brazilian franchisors and the choice of the governance mode abroad (Model 1.2). While the franchisors' business know-how by itself did not influence the choice of international governance mode (Model 1 in Table 3), it became significant in conjunction with environmental uncertainty. Thus, more uncertaint environments increased the impact of business know-how on the franchisor's probability of choosing higher control modes.

Interactive effects between environmental and behavioral uncertainty. The results for the interaction between environmental uncertainty and behavioral uncertainty contradicted the expected result (Model 1.4). While as the literature suggested, the isolated effects of greater environmental uncertainty and greater behavioral uncertainty had a negative impact on the probability of high control modes (Combs et al., 2004; Fladmoe-Lindquist and Jacque, 1995; Grewal et al., 2011; Jell-Ojobor and Windsperger, 2014), the interactive effect of environmental uncertainty mitigated the negative impact of behavioral uncertainty on the probability of choosing high control modes. A possible explanation for this result is related to the high costs that result from the interaction between behavioral uncertainty and environmental uncertainty. More uncertain environments might increase behavioral uncertainty to the point that contractual agreements such as franchising are not able to mitigate the costs. The interaction may increase franchisee-related agency costs as well, making high control modes or dual distribution preferable (Erramilli et al., 2002; Lévesque et al., 2018; Michael, 2000).

#### 6. Discussion

The empirical analysis reveals three significant findings. First, the environmental distance between the home and host countries does not impact the Brazilian franchisors' choice of international governance modes. The environmental uncertainty of the host countries is a factor that the Brazilian franchisors in this sample consider for their choice of the governance mode, despite the home country's conditions and its relative environmental distance to the host countries.

The greater the similarity between Brazil and the environments of other emerging and developing economies, which was supposed to be an advantage for Brazilian franchisors to deal with the transaction costs, does not increase the likelihood of high control modes.

#### Table 5

Logistic regression results - interaction effects.

Dependent variable: high control mode					
	Model 1.1	Model 1.2	Model 1.3	Model 1.4	
Environmental uncertainty	-1.54684***	-1.55731***	-1.46348***	-1.39213***	
	(0.47351)	(0.35408)	(0.30566)	(0.27014)	
Behavioral uncertainty	-0.76310**	$-0.81586^{**}$	-0.75621**	-0.52493	
-	(0.35286)	(0.39773)	(0.35272)	(0.33976)	
System-specific: Brand	1.20529	1.19972*	1.16171	1.12084	
	(0.74113)	(0.72070)	(0.75551)	(0.75449)	
System-specific: Know-how	0.27558	0.63041*	0.27489	0.22477	
	(0.33744)	(0.32227)	(0.33859)	(0.33847)	
Financial	0.50268*	0.50294*	0.56909*	0.46682*	
	(0.28905)	(0.30028)	(0.32928)	(0.28325)	
Size	0.46086*	0.52421**	0.43832*	0.44628*	
	(0.25587)	(0.26187)	(0.26350)	(0.25434)	
Age	-1.39411**	-1.40261**	$-1.39542^{**}$	-1.32956***	
	(0.54837)	(0.58278)	(0.54920)	(0.50593)	
Env_unc:brand	0.10447				
	(0.36841)				
Env_unc:know-how		0.43354*			
		(0.23063)			
Env_unc:financ			0.10925		
			(0.15205)		
Env_unc:behav				0.45432*	
				(0.26971)	
Constant	-2.39765***	-2.53748***	-2.37302***	-2.12236***	
	(0.75328)	(0.71848)	(0.79866)	(0.81843)	
Observations	267	267	267	267	
Sector fixed effects	Yes	Yes	Yes	Yes	
Log likelihood	-84.47139	-83.12403	-84.40116	-83.46998	
Overall Chi-square	91.482***	94.176***	91.622***	93.484***	
Nagelkerke R <sup>2</sup>	0.4657	0.4771	0.4663	0.4742	

Note: The dependent variable high control mode equals 1 if firm *i* chooses to enter the host country market with a high control mode, and zero otherwise. All of the non-binary independent variables are standardized by subtracting the mean and dividing by the standard deviation. Heteroskedasticity-robust standard errors (clustered at the firm level and operationalized by the country level) are shown in parentheses. Coefficients are statistically significant at the \*p < 0.1; \*\*p < 0.05; \*\*\*p < 0.01 levels.

Instead, these franchisors are more likely to choose low control modes when the host countries are emerging or developing countries. These findings contradict the idea that being used to high levels of uncertainty affects the behavior of firms headquartered in developing and emerging countries towards these host countries (Cuervo-Cazurra, 2012; Cuervo-Cazurra and Genc, 2008). On the contrary, Brazilian franchisors consider the host countries' characteristics, such as better environmental conditions (Lanfranchi et al., 2020; Melo et al., 2015), when making these decisions, as one can see by the high number of Brazilian franchisors operating in the United States (Aguiar et al., 2017; Cantoni et al., 2019).

Second, three main factors influence Brazilian franchisors' decisions regarding international governance modes—the environmental uncertainty of the host countries and their behavioral uncertainty, derived from the transaction cost and agency theories, and the franchisors' financial assets, derived from the resource-based and organizational capability theories. Greater environmental and behavioral uncertainties increase the tendency of franchisors to choose low control modes abroad. In host countries with greater environmental uncertainty, franchisors will be more inclined to allocate decision rights to local partners.

The same reasoning applies to distant countries with greater behavioral uncertainty. By choosing lower control modes, franchisors shift the costs of moral hazards and poor choices to the local partner. These results support de Almeida et al. (2018), that identified a preference of three Brazilian franchisors for master franchising agreements in emerging host markets: Igui, a swimming pool manufacturer operating in Mexico; Chilli Beans, that produces sunglasses and accessories and have a master franchisee in Colombia; and Young Free (fictitious name), a textile company operating in the United Arab Emirates. Brazilian firms prefer master franchising arrangements in risky locations.

On the other hand, better financial capability to fund the expansion increases the likelihood of franchisors choosing high control modes. These findings are aligned with previous empirical research that analyzed the choice of governance mode in other contexts (Castrogiovanni et al., 2006; Fladmoe-Lindquist and Jacque, 1995; Hussain et al., 2018; Jell-Ojobor and Windsperger, 2017).

Third, the environmental uncertainty of the host countries has a significant interactive effect on the Brazilian franchisors' business know-how and influences their choice of international governance mode. When the environmental uncertainty of host markets is greater, Brazilian franchisors' business know-how becomes more relevant in deciding to choose a high control governance mode. Thus, the ability to exploit their specific resource-based advantages increases in more uncertain environments (Brouthers et al., 2008; Hennart and Slangen, 2015). Additionally, environmental uncertainty has a significant interactive effect with behavioral uncertainty, mitigating the negative impact of the latter on the probability of choosing high control modes. In other words, the interaction may intensify the franchisee-related agency costs, increasing the likelihood of high control modes.

#### 7. Conclusions

In this article, we tested the multi-theoretical model of the choice of international franchising governance mode in the Brazilian context. We discussed to what extent it can explain the decisions of firms headquartered in an emerging market. The findings from the empirical analysis revealed that the similarity between the home and host countries' environment does not influence the decisions of Brazilian franchisors. Moreover, the environmental uncertainty of host markets and their behavioral uncertainty have a negative impact on their tendency to choose high control modes in international markets. In contrast, their financial assets have a positive effect on this tendency. Thus, from the analysis of the Brazilian franchisors, it is possible to conclude that the traditional international franchising literature can explain the choices of international governance modes of franchisors from an emerging market. We also observed that more uncertain environments increase the impact of business know-how and mitigate the negative impact of behavioral uncertainty on the probability of choosing high control modes.

Our results contribute to the international franchising literature by verifying if the integrative model of the choice of international franchising governance mode needs to be extended to the emerging market context (Cuervo-Cazurra, 2012; Ramamurti, 2012). We found that, besides explaining the decisions of developed countries franchisors (Jell-Ojobor and Windsperger, 2017), the theoretical model also has explanatory power to understand the choices of franchisors from an emerging market.

It also contributes by providing a statistical test to complement the theoretical evidence, case analysis, and partial results provided by previous research (Jell-Ojobor and Windsperger, 2014, 2017). Moreover, it follows the insights of Dant and Grünhagen (2014), Jell-Ojobor and Windsperger (2017), and Combs et al. (2004) by investigating the choice of governance mode in different markets, in this case, from the Brazilian franchisors' standpoint. This study also has practical implications in that it provides franchisors seeking international expansion with the variables they should consider when choosing an international governance mode.

#### 7.1. Limitations and future developments

One limitation of our study is the lack of data for other variables suggested by Jell-Ojobor and Windsperger (2014) in their multitheoretical model, namely, local market assets, transaction-specific investments, and intangible assets. Future research can improve the model by collecting data to test the effects of these three other variables — for instance, information about international partners' host market knowledge, investments, and absorptive capacity. Besides, we acknowledge that geographic distance is not the only factor that affects behavioral uncertainty. Other aspects, such as business attitudes, work ethics, qualification level of local staff, psychic distance, and language, are relevant (Burton et al., 2000; Jell-Ojobor and Windsperger, 2017). Also, the time of experience abroad could be included as an additional control variable in future studies. Unfortunately, information about these factors was not available for all the 55 host countries and 108 franchisors in the sample, and access to data about franchisors' international partners is a challenge (Windsperger and Dant, 2006).

Another limitation is related to the impact of the home country's effects. We verified the effects of the similarities between the

home and host countries by using the composite measure of the distance between the home country (Brazil) and the host countries, as well as the IMF's classification of countries as emerging and developing economies or advanced economies. In order to determine whether the home country has a direct effect on the choice of international governance mode, we need a sample of different home countries. We also recognize that the classifications of emerging and developing economies do not consider the similarities and differences between countries. Thus, we suggest future studies that investigate other home countries in emerging and developing economies.

#### Funding

This work was supported in part by the Coordenação de Aperfeiçoamento de Pessoal de Nível Superior - Brasil (CAPES) - Finance Code 001.

#### References

- ABF, 2020. Números do Franchising mostrando o desempenho do setor [WWW Document]. ABF. URL https://www.abf.com.br/numeros-do-franchising/ (accessed 2.19.20).
- Aguiar, H. de S., Consoni, F.L., Bernardes, R.C., 2014. Estratégia de internacionalização conduzida: um estudo em redes de franquias brasileiras. Revista Eletrônica de Ciência Administrativa 13, 114–131. https://doi.org/10.21529/RECADM.2014002.
- Aguiar, H. de S., Luppe, M.R., Nascimento, P.T. de S., 2017. Brazilian franchisor: entry and operation of internationalized franchise. Internext 12, 16–30. doi:10. 18568/1980-4865.12316-30.
- Alon, I., 2006a. Market conditions favoring master international franchising. Multinatl. Bus. Rev. 14, 67–82. https://doi.org/10.1108/1525383X200600009.
- Alon, I., 2006b. When Is Master International Franchising a Preferred Mode of Entry?, in: Alon, I. (Ed.), Service Franchising: A Global Perspective. Springer US, Boston, MA, pp. 103–119. doi:https://doi.org/10.1007/0-387-28256-4\_7.
- Alon, I., 2000. The organizational determinants of master international franchising. J. Bus. Entrep. 12, 1.
- Alon, I., Anderson, J., Munim, Z.H., Ho, A., 2018. A review of the internationalization of Chinese enterprises. Asia Pacific Journal of Management 35 (3), 573–605. https://doi.org/10.1007/s10490-018-9597-5.
- Alon, I., Child, J., Li, S., McIntyre, J.R., 2011. Globalization of Chinese firms theoretical universalism or particularism. Manag. Organ. Rev. 7, 191–200. https://doi.org/10.1111/j.1740-8784.2011.00234.x.
- Alon, I., McKee, D.L., 1999. The internationalization of professional business service franchises. J. Consum. Mark. 16, 74–85. https://doi.org/10.1108/ 07363769910250787
- Alon, I., Ni, L., Wang, Y., 2012. Examining the determinants of hotel chain expansion through international franchising. Int. J. Hosp. Manag. 31, 379–386. https://doi.org/10.1016/j.ijhm.2011.06.009.
- Alon, I., Welsh, D.H.B., Falbe, C.M., 2010. Franchising in Emerging Markets, in: Alon, I. (Ed.), Franchising Globally: Innovation, Learning and Imitation. Palgrave Macmillan UK, London, pp. 11–35. doi:https://doi.org/10.1057/9780230289857\_2.
- Alon, I., Yeheskel, O., Lerner, M., Zhang, W., 2013. Internationalization of Chinese entrepreneurial firms. Thunderbird Int. Bus. Rev. 55, 495–512. https://doi.org/ 10.1002/tie.21566.
- Ang, S.H., Michailova, S., 2007. Emerging economies firms' expansion in the European Union: acquisitions vs. alliances. Eur. J. Int. Manag. 1, 315–328. https://doi.org/10.1504/EJIM.2007.015654.
- Anil, I., Tatoglu, E., Ozkasap, G., 2014. Ownership and market entry mode choices of emerging country multinationals in a transition country: evidence from Turkish multinationals in Romania. Journal for East European Management Studies 19, 413–452. https://doi.org/10.5771/0949-6181-2014-4-413.
- Baena, V., 2012. Market conditions driving international franchising in emerging markets. Int. J. Emerg. Mark. 7, 49–71. https://doi.org/10.1108/ 17468801211197879.
- Baena, V., 2013. Insights on International Franchising: Entry Mode Decision. Latin American Business Review 14 (1), 1–27. https://doi.org/10.1080/ 10978526.2013.780453.
- Barkema, H.G., Vermeulen, F., 1997. What differences in the cultural backgrounds of partners are detrimental for international joint ventures? J. Int. Bus. Stud. 28, 845–864.
- Barthélemy, J., 2008. Opportunism, knowledge, and the performance of franchise chains. Strateg. Manag. J. 29, 1451–1463.
- Baum, C.F., 2006. An Introduction to Modern Econometrics Using Stata. Stata Press.
- Benito, G.R.G., Petersen, B., Welch, L.S., 2009. Towards more realistic conceptualisations of foreign operation modes. J. Int. Bus. Stud. 40, 1455–1470. Beugelsdijk, S., Ambos, B., Nell, P.C., 2018. Conceptualizing and measuring distance in international business research: recurring questions and best practice
- guidelines. J. Int. Bus. Stud. 49, 1113–1137. https://doi.org/10.1057/s41267-018-0182-4.
- Brouthers, K.D., Brouthers, L.E., Werner, S., 2008. Resource-based advantages in an international context. J. Manag. 34, 189–217. https://doi.org/10.1177/ 0149206307312508.
- Brouthers, K.D., Hennart, J.-F., 2007. Boundaries of the firm: insights from international entry mode research. J. Manag. 33, 395–425. https://doi.org/10.1177/0149206307300817.
- Brown, J.R., Dev, C.S., Zhou, Z., 2003. Broadening the foreign market entry mode decision: separating ownership and control. J. Int. Bus. Stud. 34, 473–488.
- Burton, F., Cross, A.R., Rhodes, M., 2000. Foreign market servicing strategies of UK franchisors: an empirical enquiry from a transactions cost perspective. Manag. Int. Rev. 40, 373.

Cameron, A.C., Trivedi, P.K., 2005. Microeconometrics: Methods and Applications. Cambridge University Press.

- Cantoni, A.D., Rocha, T.V., Galhanone, R.F., Righetti, M.L., 2019. International marketing strategies adopted in the franchise internationalization process: a multiple case study in the apparel industry. REMark 18, 19.
- Castrogiovanni, G.J., Combs, J.G., Justis, R.T., 2006. Resource scarcity and agency theory predictions concerning the continued use of franchising in multi-outlet networks. J. Small Bus. Manag. 44, 27–44. https://doi.org/10.1111/j.1540-627X.2006.00152.x.

Cavusgil, S.T., Knight, G., Riesenberger, J., 2017. International Business: The New Realities. Pearson.

- Chiao, Y.-C., Lo, F.-Y., Yu, C.-M., 2010. Choosing between wholly-owned subsidiaries and joint ventures of MNCs from an emerging market. Int. Mark. Rev. 27, 338–365. https://doi.org/10.1108/02651331011047998.
- Combs, J.G., David J. Ketchen, Short, JC, 2011. Franchising Research: Major Milestones, New Directions, and its Future within Entrepreneurship: Entrepreneurship Theory and Practice.
- Combs, J.G., Michael, S.C., Castrogiovanni, G.J., 2004. Franchising: a review and avenues to greater theoretical diversity. J. Manag. 30, 907–931. https://doi.org/ 10.1016/j.jm.2004.06.006.

Contractor, F.J., Kundu, S.K., 1998a. Franchising versus company-run operations: modal choice in the global hotel sector. J. Int. Mark. 6, 28–53.

- Contractor, F.J., Kundu, S.K., 1998b. Modal choice in a world of alliances: analyzing organizational forms in the international hotel sector. J. Int. Bus. Stud. 29, 325–356. https://doi.org/10.1057/palgrave.jibs.8490039.
- Cuervo-Cazurra, A., 2012. Extending theory by analyzing developing country multinational companies: solving the goldilocks debate. Glob. Strateg. J. 2, 153–167. https://doi.org/10.1111/j.2042-5805.2012.01039.x.

- Cuervo-Cazurra, A., Genc, M., 2008. Transforming disadvantages into advantages: developing-country MNEs in the least developed countries. J. Int. Bus. Stud. 39, 957–979. https://doi.org/10.1057/palgrave.jibs.8400390.
- Cuervo-Cazurra, A., Montoya, M.A., 2018. Conclusions: upgrading capabilities to build an emerging market multinational. In: Mexican Multinationals: Building Multinationals in Emerging Markets. Cambridge University Press, New York.
- Dant, R.P., Grünhagen, M., 2014. International franchising research: some thoughts on the what, where, when, and how. J. Mark. Channels 21, 124–132. https://doi.org/10.1080/1046669X.2014.917012.
- de Almeida, M. de J.F., Lanfranchi, A.G., Melo, P.L. de R., 2018. Legal environment of the destination countries of internationalized Brazilian franchise chains. Review of International Business 13, 14–27.
- Dikova, D., Brouthers, K., 2016. International establishment mode choice: past, present and future. Manag. Int. Rev. 56, 489–530. https://doi.org/10.1007/s11575-015-0258-3.
- Dikova, D., Van Witteloostuijn, A., 2007. Foreign direct investment mode choice: entry and establishment modes in transition economies. J. Int. Bus. Stud. 38, 1013–1033. https://doi.org/10.1057/palgrave.jibs.8400297.
- Doh, J., Rodrigues, S., Saka-Helmhout, A., Makhija, M., 2017. International business responses to institutional voids. J. Int. Bus. Stud. 48, 293–307. https://doi.org/ 10.1057/s41267-017-0074-z.
- Doing Business [WWW Document], 2019. World Bank. URL https://www.doingbusiness.org/en/reports/global-reports/doing-business-2019 (accessed 8.7.20). Drogendijk, R., Slangen, A., 2006. Hofstede, Schwartz, or managerial perceptions? The effects of different cultural distance measures on establishment mode choices by multinational enterprises. Int. Bus. Rev. 15, 361–380. https://doi.org/10.1016/j.ibusrev.2006.05.003.
- Elango, B., 2007. Are franchisors with international operations different from those who are domestic market oriented?\*. J. Small Bus. Manag. 45, 179–193. https:// doi.org/10.1111/i.1540-627X.2007.00208.x.
- Erramilli, K., Agarwal, S., Dev, C.S., 2002. Choice between non-equity entry modes: an organizational capability perspective. J. Int. Bus. Stud. 33, 223–242. https://doi.org/10.1057/palgrave.jibs.8491014.
- Estrin, S., Meyer, K.E., Pelletier, A., 2018. Emerging economy MNEs: how does home country munificence matter? J. World Bus. 53, 514–528.
- Fladmoe-Lindquist, K., 1996. International franchising: capabilities and development. J. Bus. Ventur. 11, 419–438. https://doi.org/10.1016/0883-9026(96)00056-0. Fladmoe-Lindquist, K., Jacque, L.L., 1995. Control modes in international service operations: the propensity to franchise. Manag. Sci. 41, 1238–1249.
- Fonseca, M.R. da, Rocha, T.V., 2018. The impact of internationalization in the brand image of Brazilian franchises of the food sector. Int. J. Professional Business Review 3, 264. https://doi.org/10.26668/businessreview/2018.v3i2.98.
- Gaur, A.S., Kumar, V., Singh, D., 2014. Institutions, resources, and internationalization of emerging economy firms. J. World Bus. 49, 12–20. https://doi.org/ 10.1016/j.jwb.2013.04.002.
- Ghantous, N., Das, S.S., 2018. International franchising and performance: a resource-based perspective. Int. J. Retail Distrib. Manag. 46, 744–763. https://doi.org/ 10.1108/IJRDM-10-2017-0244.
- Gillis, W., Combs, J., Ketchen, J.D., 2014. Using resource-based theory to help explain plural form franchising. Entrepreneurship Theory and Practice 38, 449–472. https://doi.org/10.1111/etap.12008.
- Grewal, D., Iyer, G.R., Javalgi, R. (Raj) G., Radulovich, L., 2011. Franchise partnership and international expansion: a conceptual framework and research propositions. Entrepreneurship Theory and Practice 35, 533–557. doi:https://doi.org/10.1111/j.1540-6520.2011.00444.x.
- Guillén, M.F., García-Canal, E., 2009. The American model of the multinational firm and the "new" multinationals from emerging economies. Acad. Manag. Perspect. 23, 23–35.
- Hair, J.F., Jr., Anderson, R.E., Tatham, R.L., Black, W.C., 1995. Multivariate Data Analysis (4th Ed.): With Readings. Prentice-Hall, Inc., Upper Saddle River, NJ, USA. Hansmann, H., 1996. The Ownership of Enterprise, 1st edition. ed. Belknap Press, Cambridge.
- Hennart, J.-F., Slangen, A., 2015. Yes, we really do need more entry mode studies! A commentary on shaver. J. Int. Bus. Stud. 46, 114–122. https://doi.org/10.1057/jibs.2014.39.
- Hoffman, R., Watson, S., Kher, H., 2020. Modes of governance for market entry by international franchisors: factors affecting the choice. Management Research Review Ahead-of-Print. https://doi.org/10.1108/MRR-07-2020-0400.
- Hoffman, R.C., Munemo, J., Watson, S., 2016. International franchise expansion: the role of institutions and transaction costs. J. Int. Manag. 22, 101–114. https://doi.org/10.1016/j.intman.2016.01.003.
- Hoskisson, R.E., Eden, L., Lau, C.M., Wright, M., 2000. Strategy in emerging economies. AMJ 43, 249-267. https://doi.org/10.5465/1556394.
- Hoskisson, R.E., Wright, M., Filatotchev, I., Peng, M.W., 2013. Emerging multinationals from mid-range economies: the influence of institutions and factor markets. J. Manag. Stud. 50, 1295–1321. https://doi.org/10.1111/j.1467-6486.2012.01085.x.
- Hussain, D., Sreckovic, M., Windsperger, J., 2018. An organizational capability perspective on multi-unit franchising. Small Bus. Econ. 50, 717–727. https://doi.org/ 10.1007/s11187-017-9891-5.
- Jain, N.K., Pangarkar, N., Yuan, L., Kumar, V., 2015. Internationalization of Indian software firms through establishment of global development centers: a contingency perspective. The Multinational Business Review 23, 90–110. https://doi.org/10.1108/MBR-06-2014-0033.
- Jell-Ojobor, M., Alon, I., 2017. Determinants of master international franchising. In: Handbook of Research on Franchising. Edward Elgar Publishing Limited, Cheltenham, UK
- Jell-Ojobor, M., Windsperger, J., 2017. Determinants of the governance structure of the international franchise firm: a case study analysis in the automotive rental industry. Int. Mark. Rev. 34, 814–884. https://doi.org/10.1108/IMR-10-2015-0208.
- Jell-Ojobor, M., Windsperger, J., 2014. The choice of governance modes of international franchise firms development of an integrative model. J. Int. Manag. 20, 153–187. https://doi.org/10.1016/j.intman.2013.09.001.
- Kedia, B.L., Ackerman, D.J., Bush, D.E., Justice, R.T., 1994. Determinants of internationalization of franchise operations by US franchisors. Int. Mark. Rev. 11, 56–68. https://doi.org/10.1108/02651339410069254.
- Kedia, B.L., Reddy, R.K., Bilgili, T.V., 2015. Forging ahead with innovation: the importance of institutions and factor markets to the continued development of the BRICS. In: Emerging Markets and the Future of the BRIC Nations. Edward Elgar Publishing Limited, p. 224.
- Khanna, T., Palepu, K.G., Sinha, J., 2005. Strategies that fit emerging markets. Harv. Bus. Rev. 83, 4-19.
- Lafontaine, F., 2014. Franchising: directions for future research. Int. J. Econ. Bus. 21, 21–25. https://doi.org/10.1080/13571516.2013.864124.
- Lafontaine, F., Shaw, K.L., 2005. Targeting managerial control: evidence from franchising. RAND J. Econ. 36, 131-150.
- Lanfranchi, A., Melo, P.L. de R., Borini, F.M., Telles, R., 2020. Institutional environment and internationalization of franchise chains: a regional and global analysis. International Journal of Emerging Markets ahead-of-print. https://doi.org/10.1108/IJOEM-03-2019-0188.
- Lévesque, M., Choi, Y.R., Hsuan, J., 2018. Growth through franchises in knowledge-intensive industries: interplay of routine program and expansion mode. I E E E TRANSACTIONS ON ENGINEERING MANA 66.
- Madanoglu, M., Alon, I., Shoham, A., 2017. Push and pull factors in international franchising. Int. Mark. Rev. https://doi.org/10.1108/IMR-03-2015-0037.
- Mariz-Pérez, R., García-Álvarez, T., 2009. The internationalization strategy of Spanish indigenous franchised chains: a resource-based view\*. J. Small Bus. Manag. 47, 514–530. https://doi.org/10.1111/j.1540-627X.2009.00281.x.
- Mathews, J.A., 2006. Dragon multinationals: new players in 21st century globalization. Asia Pacific J Manage 23, 5–27. https://doi.org/10.1007/s10490-006-6113-0. Mayer, T., Zignago, S., 2011. Notes on CEPII's Distances Measures: The GeoDist Database (CEPII Working Paper No. 25). CEPII.
- Melo, P.L. de R., Borini, F.M., Ogasavara, M.H., 2019. Latin American franchise internationalization: the impact of institutional environment. Thunderbird Int. Bus. Rev. 61, 217–228. https://doi.org/10.1002/tie.21975.
- Melo, P.L.D.R., Borini, F.M., Oliveira Jr., M.D.M., Parente, R.C., Melo, P.L.D.R., Borini, F.M., Oliveira Jr., M.D.M., Parente, R.C., 2015. Internationalization of Brazilian franchise chains: a comparative study. Revista de Administração de Empresas 55, 258–272. https://doi.org/10.1590/S0034-759020150303.
- Merrilees, B., 2014. International franchising: evolution of theory and practice. J. Mark. Channels 21, 133–142. https://doi.org/10.1080/1046669X.2014.917013.

- Meyer, K.E., Estrin, S., Bhaumik, S.K., Peng, M.W., 2009. Institutions, resources, and entry strategies in emerging economies. Strateg. Manag. J. 30, 61–80. https://doi.org/10.1002/smj.720.
- Michael, S.C., 2000. The effect of organizational form on quality: the case of franchising. J. Econ. Behav. Organ. 43, 295–318. https://doi.org/10.1016/S0167-2681 (00)00125-6.
- Michailova, S., Ang, S.H., 2008. Institutional explanations of cross-border alliance modes: the case of emerging economies firms. MANAGE. INT. REV. 48, 551–576. https://doi.org/10.1007/s11575-008-0036-6.
- Moraes, W.F. de, Oliveira, B.R.B. de, Kovacs, É.P., 2008. Teorias de internacionalização e aplicação em países emergentes: uma análise crítica. Internext 1, 221–242. doi:10.18568/1980-4865.11221-242.
- Morschett, D., Schramm-Klein, H., Swoboda, B., 2010. Decades of research on market entry modes: what do we really know about external antecedents of entry mode choice? J. Int. Manag., 50 years of IB research: What have we achieved, and What have we not yet achieved? 16, 60–77. doi:https://doi.org/10.1016/j.intman. 2009.09.002.
- Mumdžiev, N., 2011. Allocation of decision rights in international franchise firms: the case of master and direct franchising. In: Tuunanen, M., Windsperger, J., Cliquet, G., Hendrikse, G. (Eds.), New Developments in the Theory of Networks: Franchising, Alliances and Cooperatives, Contributions to Management Science. Physica-Verlag HD, Heidelberg, pp. 45–58. https://doi.org/10.1007/978-3-7908-2615-9 4.
- Ni, L., Alon, I., 2010. U.S.-based fast-food restaurants: factors influencing the international expansion of franchise systems. J. Mark. Channels 17, 339–359. https:// doi.org/10.1080/1046669X.2010.512861.
- Nielsen, U.B., Hannibal, M., Larsen, N.N., 2018. Reviewing emerging markets: context, concepts and future research. Int. J. Emerg. Mark. 13, 1679–1698. https://doi. org/10.1108/IJoEM-10-2017-0416.

North, D.C., 1991. Institutions. J. Econ. Perspect. 5, 97–112. https://doi.org/10.1257/jep.5.1.97.

North, D.C., 1990. Institutions Institutional Change and Economic Performance|Political Economy. Cambridge University Press, Cambridge, UK

- Pan, Y., Tse, D.K., 2000. The hierarchical model of market entry modes. J. Int. Bus. Stud. 31, 535–554. https://doi.org/10.1057/palgrave.jibs.8490921.
- Perrigot, R., Cliquet, G., Mesbah, M., 2004. Possible applications of survival analysis in franchising research. Int. Rev. Retail Distrib. Consum. Res. 14, 129–143. https://doi.org/10.1080/0959396032000154338.
- Perrigot, R., López-Fernández, B., Eroglu, S., 2013. Intangible resources and plural form as drivers of franchise internationalization: examination within a two-country perspective. J. Small Bus. Manag. 51, 557–577. https://doi.org/10.1111/jsbm.12006.

Ramamurti, R., 2012. What is really different about emerging market multinationals? Glob. Strateg. J. 2, 41-47. https://doi.org/10.1002/gsj.1025.

- Ramamurti, R., 2009. What have we learned about emerging—market MNEs?. In: Emerging Multinationals in Emerging Markets. Cambridge University Press, p. 399. Rocha, T.V., Spers, E.E., Borini, F.M., Bretas, V.P., Melo, P.R., Ogasavara, M.H., Camargo, M.A., 2018. A marca na internacionalização das franquias brasileiras. ESPM, São Paulo
- Rosado-Serrano, A., Paul, J., Dikova, D., 2018. International franchising: a literature review and research agenda. J. Bus. Res. 85, 238–257. https://doi.org/10.1016/j. jbusres.2017.12.049.
- Rugman, A., 2010. Do We Need A New Theory To Explain Emerging Market Multinationals?.
- Sashi, C.M., Karuppur, D.P., 2002. Franchising in global markets: towards a conceptual framework. Int. Mark. Rev. 19, 499–524. https://doi.org/10.1108/ 02651330210445301.
- Shane, S.A., 1996. Why franchise companies expand overseas. J. Bus. Ventur. 11, 73-88. https://doi.org/10.1016/0883-9026(95)00110-7.
- Surdu, I., Mellahi, K., Glaister, K., 2018. Emerging market multinationals' international equity-based entry mode strategies. Int. Mark. Rev. 35, 342–359. https://doi. org/10.1108/IMR-10-2015-0228.
- Watson, A., Stanworth, J., Healeas, S., Purdy, D., Stanworth, C., 2005. Retail franchising: an intellectual capital perspective. J. Retail. Consum. Serv. 12, 25–34. https://doi.org/10.1016/j.jretconser.2004.02.001.
- Windsperger, J., 2004. Centralization of franchising networks: evidence from the Austrian franchise sector. Journal of Business Research, Mobility and Markets: Emerging Outlines of M-Commerce 57, 1361–1369. https://doi.org/10.1016/S0148-2963(03)00068-7.
- Windsperger, J., Dant, R.P., 2006. Contractibility and ownership redirection in franchising: a property rights view. J. Retail. 82, 259–272. https://doi.org/10.1016/j. jretai.2006.06.001.
- Windsperger, J., Kocsis, E., Rosta, M., 2009. Exploring the relationship between decision and ownership rights in joint ventures: evidence from Hungarian joint ventures. International Studies of Management & Organization 39, 43–59.
- Windsperger, J., Yurdakul, A., 2007. The governance structure of franchising firms: a property rights approach. In: Cliquet, G., Tuunanen, M., Hendrikse, G., Windsperger, J. (Eds.), Economics and Management of Networks: Franchising, Strategic Alliances, and Cooperatives, Contributions to Management Science. Physica-Verlag HD, Heidelberg, pp. 69–95. https://doi.org/10.1007/978-3-7908-1758-4 5.
- Wright, M., Filatotchev, I., Hoskisson, R.E., Peng, M.W., 2005. Strategy research in emerging economies: challenging the conventional wisdom. J. Manag. Stud. 42, 1–33. https://doi.org/10.1111/j.1467-6486.2005.00487.x.